EXECUTION VERISION

SECURITIES PURCHASE AGREEMENT

By and Among

BDT CF ACQUISITION VEHICLE, LLC

and

COLFAX CORPORATION

Dated as of September 12, 2011

SECURITIES PURCHASE AGREEMENT, dated as of September 12, 2011 (together with the schedules and exhibits hereto, the "Agreement"), by and among Colfax Corporation, a Delaware corporation (the "Company") and BDT CF Acquisition Vehicle, LLC, a Delaware limited liability company (the "Investor").

BDT Capital Partners Fund I, L.P., a Delaware limited partnership ("BDT Main"), and BDT Capital Partners Fund I-A, L.P., a Delaware limited partnership ("BDT Parallel") are parties to this Agreement solely for purposes of Section 5.11. BDT Main and BDT Parallel are referred to herein collectively as the "BDT Fund". Mitchell P. Rales and Steven M. Rales (the "Designated Stockholders") are parties to this Agreement solely for purposes of Section 5.12.

RECITALS

- The Investor desires to purchase from the Company, and the Company A. desires to issue and sell to the Investor, against payment of the Investment Amount (as defined below) and pursuant to the terms and conditions set forth in this Agreement, certain shares of (a) the Company's preferred stock, par value \$0.001 per share, designated as Series A Perpetual Convertible Preferred Stock ("Preferred Shares"), having the voting and other powers, preferences and relative, participating, optional or other rights, and the qualifications, limitations and restrictions as specified in the Amended and Restated Certificate of Incorporation of the Company to be adopted on or before the Closing Date in the form attached hereto as Exhibit A (the "Amended and Restated Charter") and the certificate of designations establishing the Preferred Shares to be adopted pursuant to the Amended and Restated Charter on or before the Closing Date in the form attached hereto as Exhibit B (the "Certificate of Designations"), and (b) the Company's common stock, par value \$0.001 per share ("Common Stock"). The Preferred Shares to be purchased by the Investor pursuant to this Agreement are referred to herein as the "Purchased Preferred Shares"; and the shares of Common Stock to be purchased by the Investor pursuant to this Agreement are referred to herein as the "Purchased Common Shares". The Purchased Preferred Shares and the Purchased Common Shares are referred to herein collectively as the "Purchased Shares". The sale and purchase of the Purchased Shares and the other transactions contemplated by this Agreement and the Ancillary Agreements are referred to herein collectively as the "Transaction".
- B. On or prior to the date hereof, certain parties have entered into equity commitment letters (the "Commitment Letters") with the Investor, pursuant to which such parties (the "Co-Investors") have committed, upon the terms and conditions set forth therein, to make aggregate capital contributions of \$290,000,000 to the Investor in order to finance in part the Investor's payment of the Investment Amount pursuant to this Agreement. In addition, the BDT Fund has covenanted and agreed, subject to the terms and conditions set forth in this Agreement, to make an aggregate capital contribution of \$390,000,000 (\$30,000,000 of which will be contributed by BDT Main pursuant to the terms of an equity commitment letter (the "BDT Main Equity Commitment Letter")) to the Investor in order to finance in part the Investor's payment of the Investment Amount pursuant to this Agreement. The aggregate capital commitments of the BDT Fund and the Co-Investors pursuant to this Agreement and the Commitment Letters, respectively, are equal to \$680,000,000.

- C. Concurrent with the execution and delivery of this Agreement, and in connection with the Acquisition (as defined below), the Company is entering into a credit agreement (the "Credit Agreement"), with Deutsche Bank AG New York Branch ("Deutsche Bank"), a syndicate of other financial institutions as lenders (the "Lenders"), and Deutsche Bank Securities, Inc., with respect to credit facilities to be provided to the Company and certain of its subsidiaries by Deutsche Bank and the Lenders.
- D. Concurrent with the execution and delivery of this Agreement, and in connection with the Acquisition, (i) the Company is entering into purchase agreements with Mitchell P. Rales, Steven M. Rales and Markel Corporation, pursuant to which they shall agree to invest an aggregate amount of \$125,000,000 in newly issued shares of Common Stock (the "Designated Stockholder Investment"), and (ii) Mitchell P. Rales and Stephen M. Rales are entering into a voting agreement, pursuant to which they are agreeing to vote all shares of Common Stock owned or controlled by them in favor of the approval of the Transaction in any meetings of the stockholders of the Company called for such purpose.
- E. In accordance with this Agreement and the Credit Agreement, the Company shall use the proceeds of the Transaction and the Designated Stockholder Investment and a portion of the proceeds of the Company's borrowings under the Credit Agreement to finance the acquisition (the "Acquisition") by the Company of Charter International plc (the "Target") pursuant to an Offer or a Scheme of Arrangement (each as defined below). The material terms and provisions of the Acquisition are set forth in the form of announcement attached hereto as **Exhibit C** (the "Rule 2.5 Announcement"), to be published by the Company in accordance with this Agreement pursuant to Rule 2.5 of the City Code.

NOW, THEREFORE, in consideration of the foregoing and the promises and representations, warranties, covenants and agreements contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and intending to be legally bound hereby, the parties hereto hereby agree as follows:

ARTICLE 1

Defined Terms

As used in this Agreement, the following terms shall have the following meanings:

"Acquisition" has the meaning assigned to such term in Recital E.

"Action" means any suit, action, proceeding (including any compliance, enforcement or disciplinary proceeding), arbitration, formal or informal inquiry, inspection, investigation or formal order of investigation or complaint, or other litigation of any kind, in each case whether civil, criminal or administrative, at law or in equity.

"Affiliate" of any person means another person that, directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with, such first person. A person shall be deemed to control another person if such first person

possesses, directly or indirectly, the power to direct, or cause the direction of, the management and policies of such other person, whether through the ownership of voting securities, by contract or otherwise.

"Agreement" has the meaning assigned to such term in the Preamble.

"Amended and Restated Charter" has the meaning assigned to such term in Recital A.

"Ancillary Agreements" means the Registration Rights Agreement and the certificates to be delivered by the parties pursuant to Sections 2.5(a)(ii), 2.5(a)(iii), and 2.5(b)(ii).

"Announcement Date" has the meaning assigned to such term in Section 2.3.3(a).

"Antitrust Laws" has the meaning assigned to such term in Section 5.7.

"BDT Fund" has the meaning assigned to such term in the Preamble to this Agreement.

"BDT Main" has the meaning assigned to such term in the Preamble to this Agreement.

"BDT Main Equity Commitment Letter" has the meaning assigned to such term in Recital B.

"BDT Parallel" has the meaning assigned to such term in the Preamble to this Agreement.

"Board of Directors" has the meaning assigned to such term in Section 3.2(a).

"Business Day" means any day except Saturday, Sunday or a day that is a public holiday in England, Jersey or in the United States of America.

"Bylaws" means the Amended and Restated Bylaws of the Company, adopted effective as of May 13, 2008, in the form provided to the Investor as an annex to the secretary's certificate delivered by the Company pursuant to Section 2.5(a)(iii).

"Cash Confirmation Advisor" means Deutsche Bank AG, London Branch.

"Certificate of Designations" has the meaning assigned to such term in Recital A.

"Charter" means the Amended and Restated Certificate of Incorporation of the Company, as filed with the Secretary of State of the State of Delaware on May 13, 2008, in the form provided to the Investor as an annex to the secretary's certificate delivered by the Company pursuant to Section 2.5(a)(iii).

"City Code" means the City Code on Takeovers and Mergers.

"Closing" has the meaning assigned to such term in Section 2.4.

- "Closing Date" has the meaning assigned to such term in Section 2.4.
- "Code" means the Internal Revenue Code of 1986, as amended.
- "Co-Investors" has the meaning assigned to such term in Recital B.
- "Common Share Investment Amount" has the meaning assigned to such term in Section 2.2.
- "Common Share Lock Up Period" has the meaning assigned to such term in Section 5.8(a).
- "Common Share Purchase Price" has the meaning assigned to such term in Section 2.3.3(b).
 - "Common Stock" has the meaning assigned to such term in Recital A.
- "Common Stock Equivalents" means any options, warrants, convertible securities or other securities, instruments or rights convertible into or exchangeable or exercisable for shares of Common Stock.
- "Companies Act" means the Companies (Jersey) Law 1991 as amended from time to time.
- "Company Disclosure Schedules" has the meaning assigned to such term in the introductory statement to <u>ARTICLE 3</u>.
- "Competitor" means a Person that competes with the Company or any of its subsidiaries in a material way in any line of business accounting for twenty (20%) or more of the consolidated revenues of the Company and its subsidiaries, as reflected in the Company's consolidated financial statements filed with the SEC for the calendar quarter immediately preceding the date of determination.
- "Contract" means any contract, agreement, lease, purchase order, license, mortgage, indenture, supplemental indenture, line of credit, note, bond, loan, credit agreement, capital lease, sale/leaseback arrangement, concession agreement, franchise agreement or other instrument, including all amendments, supplements, exhibits and attachments thereto.
- "Court Meeting" means a meeting convened by the High Court of Jersey between the owners of the Target Shares to seek their approval of the Scheme of Arrangement.
 - "Credit Agreement" has the meaning assigned to such term in Recital C.
- "<u>Designated Stockholder</u>" has the meaning assigned to such term in the Preamble to this Agreement.
- "<u>Designated Stockholder Investment</u>" has the meaning assigned to such term in Recital D.

"Encumbrance" means any lien, encumbrance, security interest, pledge, mortgage, hypothecation, charge, restriction on transfer of title, adverse claim, title retention agreement of any nature or kind, or similar claim or right. The term "Encumbrance" shall not include any restrictions on transfer applicable to the Purchased Shares under applicable securities laws.

"Exchange Act" means the Securities Exchange Act of 1934, as amended.

"Financial Statements" has the meaning assigned to such term in Section 3.6(b).

"<u>Fund Commitment Amount</u>" has the meaning assigned to such term in <u>Section</u> 5.11.

"GAAP" means generally accepted accounting principles in the United States of America as in effect from time to time.

"General Meeting" means an extraordinary general meeting of the Target convened at the direction of the High Court of Jersey to cause, and if thought fit, approve one or more resolutions to reduce the ordinary share capital of the Target and approve a Scheme of Arrangement.

"Governmental Authorizations" means, collectively, all applicable consents, approvals, permits, orders, authorizations, licenses and registrations, given or otherwise made available by or under the authority of Governmental Entities or pursuant to the requirements of any Applicable Law.

"Governmental Entity" means any domestic or foreign, transnational, national, Federal, state, municipal or local government, or any other domestic or foreign governmental, regulatory or administrative authority, or any agency, board, department, commission, court, tribunal or instrumentality thereof.

"Guarantor" means a financial institution that has delivered, for the benefit of the Investor, a performance bond (each, a "Guarantee") of the liabilities of the BDT Fund or a Co-Investor, as applicable, for such entity's committed portion of the Investment Amount.

"HSR Act" means Hart Scott Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations promulgated thereunder.

"including" means including, without limitation.

"Investor" has the meaning assigned to such term in the Preamble.

"Investment Amount" has the meaning assigned to such term in Section 2.2.

"Judgment" means any applicable judgment, order or decree of any Governmental Entity.

"knowledge of the Company," "to the Company's knowledge" or other references to the "knowledge" of the Company mean the knowledge of the particular fact in question by any officer or senior executive of the Company, assuming exercise of reasonable diligence and inquiry.

"Lenders" has the meaning assigned to such term in Recital C.

"Major Representation" means each representation and warranty set forth in Sections 3.1, 3.2 and 3.3 of this Agreement

"Major Undertaking" means each covenant set forth in Section 2.5(a), Section 5.5 and Section 5.6 of this Agreement (other than those set forth in Sections 5.6(g), 5.6(h), 5.6(i), 5.6(j) and 5.6(k)).

"Material Adverse Effect" means any material adverse effect on (i) the business, results of operations or financial condition of the Company and its subsidiaries, taken as a whole (other than (a) any change, event, occurrence or development that generally affects the industry in which the Company and its subsidiaries operate and does not disproportionately affect (relative to other industry participants) the Company and its subsidiaries, (b) any change, event, occurrence or development to the extent attributable to conditions generally affecting (I) the industries in which the Company participates that does not have a materially disproportionate effect (relative to other industry participants) on the Company and its subsidiaries, (II) the U.S. economy as a whole, or (III) the equity capital markets generally, and (c) any change, event, occurrence or development that results from any action taken by the Company at the request of the Investor or as required by the terms of this Agreement) or (ii) the ability of the Company to perform its obligations hereunder or under the Registration Rights Agreement.

"Material Contract" means any Contract of a type described in Item 6.01 of Regulation S-K.

"Offer" means a public offer by the Company in accordance with the City Code and the provisions of the Companies Act to acquire all of the Target Shares not owned, held, or agreed to be acquired by the Company.

"Offer Document" means an offer document dispatched to shareholders of the Target setting out in full the terms and conditions of the Offer.

"Permitted Transferee" means the Co-Investors, any direct or indirect partners or members of the Investor or the BDT Fund, or any Affiliates of any of the foregoing (including any pension fund maintained by or for the benefit of any of the foregoing), in each case, other than a Competitor. For purposes of clarity, any other Third Party to whom any shares of Common Stock, Purchased Preferred Shares or Underlying Shares are transferred following the lapse of the applicable restrictions set forth in Section 5.8 shall not be a Permitted Transferee.

"person" means any individual, firm, corporation, partnership, limited liability company, trust, joint venture, incorporated or unincorporated association, Governmental Entity or other entity.

- "Preferred Shares" has the meaning assigned to such term in Recital A.
- "Preferred Share Investment Amount" has the meaning assigned to such term in Section 2.2.
- "Preferred Share Purchase Price" has the meaning assigned to such term in Section 2.3.3(c).
 - "Proposed Transferee" has the meaning assigned to such term in Section 5.12(a).
 - "Purchased Common Shares" has the meaning assigned to such term in Recital A.
 - "Purchased Preferred Shares" has the meaning assigned to such term in Recital A.
 - "Purchased Shares" has the meaning assigned to such term in Recital A.
- "Recent Balance Sheet" means the most recent consolidated balance sheet of the Company and its consolidated subsidiaries included in the Financial Statements.
- "Reduction Court Order" means the order of the High Court of Jersey confirming the reduction of the capital of the Target provided for by the Scheme of Arrangement.
- "Registration Rights Agreement" means the Registration Rights Agreement, in the form attached hereto as **Exhibit D**, to be executed on the Closing Date by and among, the Company, the Investor, and the other parties specified therein.
 - "Rule 2.5 Announcement" has the meaning assigned to such term in Recital E.
- "Scheme of Arrangement" means a scheme of arrangement effected pursuant to Part 18A of the Companies Act under which the Target Shares will be cancelled (or transferred) and the Purchaser will become the holder of new shares issued in place of such cancelled Target Shares (or, as the case may be, the holder of such transferred Target Shares).
- "Scheme Circular" means a circular dispatched by the Target to holders of the Target Shares setting out in full the terms and conditions of a Scheme of Arrangement and convening a General Meeting and a Court Meeting.
- "Scheme Court Order" means the order of the High Court of Jersey sanctioning the Scheme of Arrangement.
 - "SEC" means the Securities and Exchange Commission.
 - "SEC Reports" has the meaning assigned to such term in Section 3.6(a).
 - "Securities Act" means the Securities Act of 1933, as amended.
 - "SOX" means the Sarbanes-Oxley Act of 2002.

"subsidiary" of any person means any person the accounts of which would be consolidated with and into those of the first person in such first person's consolidated financial statements if such financial statement were prepared in accordance with GAAP.

"Tag-Along Transaction" has the meaning assigned to such term in Section 5.12(a).

"Tag-Along Offer" has the meaning assigned to such term in Section 5.12(b).

"Takeover Panel" means the Panel on Takeovers and Mergers.

"Target" has the meaning assigned to such term in Recital E.

"<u>Target Shares</u>" means the ordinary shares of the Target subject of an Offer, or as the case may be, a Scheme of Arrangement.

"Tax" means any foreign, Federal, state or local income, sales and use, excise, franchise, real and personal property, gross receipt, capital stock, license, stamp, premium, windfall profit, environmental, transfer, alternative or add-on minimum, production, business and occupation, disability, estimated, employment, payroll, severance or withholding tax, value-added tax or other tax, duty, fee, impost, levy, assessment or charge imposed by any taxing authority, and any interest or penalties and other additions to tax related thereto.

"<u>Tax Returns</u>" means any return, report, claim for refund, declaration, information return or other document required to be filed with any Tax authority with respect to Taxes, including any schedule or attachment thereto, and including any amendments thereof.

"Third Party" means any person other than the Company, the Investor, or any of their respective subsidiaries or Affiliates.

"Transaction" has the meaning assigned to such term in Recital A.

"Transfer Date" has the meaning set forth in Section 5.12(b).

"<u>Underlying Shares</u>" means the shares of the Company's Common Stock issuable upon conversion of the Purchased Preferred Shares in accordance with the terms and provisions thereof.

"<u>Voting Debt</u>" means bonds, debentures, notes or other debt securities having the right to vote (or convertible into, or exchangeable for, securities having the right to vote) as to matters on which holders of the Company's Common Stock may vote.

"<u>Voting Stock</u>" of any person means securities having the right to vote generally in any election of directors or comparable governing persons of any such person.

ARTICLE 2

Issuance and Sale of Purchased Shares

- 2.1 Issuance and Sale of the Purchased Shares. On the terms and subject to the conditions set forth in this Agreement, at the Closing (as defined below), the Company shall issue, sell and deliver to the Investor, and the Investor shall purchase from the Company, the Purchased Preferred Shares and the Purchased Common Shares, in each case in certificated form and free and clear of all Encumbrances. The number of Preferred Shares and shares of Common Stock comprising the Purchased Preferred Shares and Purchased Common Shares to be issued and delivered to the Investor at the Closing shall be determined in accordance with Section 2.3.
- Consideration. In consideration for the purchase of the Purchased Shares, 2.2 the Investor shall pay to the Company at the Closing all of the amounts contributed to it by the BDT Fund pursuant to Section 5.11, all of the amounts contributed to it by BDT Main pursuant to the BDT Main Equity Commitment Letter, all of the amounts contributed to it by the Co-Investors pursuant to the Commitment Letters, all of the amounts contributed to it by a Permitted Assignee, and, if applicable, all of the amounts paid to it by the Guarantors under the applicable Guarantees which, subject to the terms of Section 5.11 and the Commitment Letters being fully complied with and/or payment being made under such Guarantees, shall total: (a) with respect to the Purchased Preferred Shares, cash in the amount of \$340,000,000 (the aggregate amount actually delivered with respect to the Purchased Preferred Shares, the "Preferred Share Investment Amount"), and (b) with respect to the Purchased Common Shares, cash in the amount of \$340,000,000 (the aggregate amount actually delivered with respect to the Purchased Common Shares, the "Common Share Investment Amount" and together with the Preferred Share Investment Amount, the "Investment Amount").

2.3 Determination of Purchased Share Amounts.

- 2.3.1 <u>Purchased Preferred Shares</u>. The number of Preferred Shares comprising the Purchased Preferred Shares to be issued and delivered to the Investor at the Closing pursuant to <u>Section 2.1</u> shall be equal to the quotient of (a) the Preferred Share Investment Amount, *divided by* (b) the Preferred Share Purchase Price (as defined in <u>Section 2.3.3</u>), rounded up to the nearest whole number of Preferred Shares, which number of Preferred Shares shall be the authorized number of Preferred Shares in the Certificate of Designations.
- 2.3.2 <u>Purchased Common Shares</u>. The number of shares of Common Stock comprising the Purchased Common Shares to be issued and delivered to the Investor at the Closing pursuant to <u>Section 2.1</u> shall be equal to the quotient of (a) the Common Share Investment Amount, *divided by* (b) the Common Share Purchase Price (as defined in <u>Section 2.3.3</u>), rounded up to the nearest whole number of shares of Common Stock.
- 2.3.3 <u>Certain Definitions</u>. As used in this <u>Section 2.3</u>, the following terms shall have the following meanings:
- (a) "Announcement Date" means the date that the Rule 2.5 Announcement is published.
 - (b) "Common Share Purchase Price" shall equal \$23.04.

- (c) "Preferred Share Purchase Price" shall equal \$24.50.
- 2.4 Closing. Subject to the satisfaction of the conditions set forth in ARTICLE 6, the closing of the sale and purchase of the Purchased Shares (the "Closing") shall take place on the date falling six (6) Business Days after the date on which an Offer becomes unconditional in all respects or, as the case may be, the date on which a Scheme of Arrangement becomes effective in accordance with its terms, at the offices of the Company unless another date or place is agreed to in writing by the Investor and the Company. The date on which the Closing occurs is referred to in this Agreement as the "Closing Date".

2.5 Deliveries.

- (a) At the Closing, the Company shall deliver to the Investor:
- (i) The Purchased Shares, validly issued and free and clear of Encumbrances, evidenced by certificates registered in the name of the Investor and duly authorized, executed and delivered on behalf of the Company, bearing the legend set forth to in Section 4.3;
- (ii) A certificate, dated as of the Closing Date, signed by an authorized officer of the Company, in the form attached hereto as **Exhibit E**, certifying that the conditions set forth in Section 6.3 have been satisfied;
- (iii) A certificate, dated as of the Closing Date, signed by the secretary of the Company, in the form attached hereto as Exhibit F, certifying as to the matters set forth therein; and
- (iv) A duly executed counterpart of the Registration Rights Agreement.
 - (b) At the Closing, the Investor shall deliver to the Company:
- (i) The Investment Amount in immediately available funds by wire transfer to such account as the Company shall specify in writing not later than three (3) Business Days prior to the Closing Date;
- (ii) A certificate, dated as of the Closing Date, signed by an authorized officer of the Investor, in form attached hereto as **Exhibit G**, certifying that the conditions set forth in <u>Sections 6.2</u> have been satisfied; and
- (iii) A duly executed counterpart of the Registration Rights Agreement.

ARTICLE 3

Representations and Warranties of the Company

Subject to the qualifications set forth in the corresponding sections of the disclosure schedules delivered by the Company to the Investor concurrently with the execution and delivery of this Agreement (the "Company Disclosure Schedules"), the Company hereby represents and warrants to the Investor, as of the date hereof and as of the Closing Date, as follows:

3.1 Entity Status. Each of the Company and its subsidiaries is duly incorporated or otherwise organized, validly existing and in good standing under the applicable laws of its governing jurisdiction and each has all requisite corporate or other power and authority to carry on its business as it is now being conducted and is duly qualified to do business in each of the jurisdictions in which the ownership, operation or leasing of its assets or the conduct of its business requires it to be so qualified, and no proceeding has been instituted in any such jurisdiction revoking, limiting or curtailing, or seeking to revoke, limit or curtail, such power, authority or qualification, in each case, except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.

3.2 Authorization; Noncontravention.

Authorization. The Company has all necessary corporate (a) power and authority to execute and deliver this Agreement and the Ancillary Agreements, to perform its obligations hereunder and thereunder, and to consummate the Transaction and the Acquisition. The Company has delivered to the Investor a true and correct copy, certified by the Company's secretary, of the resolutions of its board of directors (the "Board of Directors") authorizing (i) the execution and delivery of this Agreement, (ii) consummation of the Transaction, the Acquisition and the Designated Stockholder Investment, (iii) recommending that the Company's stockholders approve the Transaction and the Designated Stockholder Investment and the adoption of the Amended and Restated Charter and the Certificate of Designations, and (iv) waiving the restrictions set forth in Section 203 of the Delaware General Corporation Law. Such resolutions are in full force and effect, have not been amended, supplemented, revoked or superseded as of the date hereof and are the only resolutions of the Board of Directors pertaining to the authorization, execution and delivery of this Agreement and the Ancillary Agreements and consummation of the Transaction. The execution, delivery and performance of this Agreement and the Ancillary Agreements and the consummation by the Company of the Transaction (including the issuance (or reservation for issuance), sale and delivery of the Purchased Shares and the Underlying Shares and any redemptions of the Purchased Preferred Shares pursuant to the terms thereof) and the Acquisition, have been duly and validly authorized by all necessary corporate action (except the stockholder actions contemplated by Section 5.1), and no other corporate proceedings on the part of the Company or its subsidiaries or (except the stockholder actions contemplated by Section 5.1) vote of holders of any class or series of capital stock of the Company or its subsidiaries is necessary to authorize this Agreement or the Ancillary Agreements or to consummate the Transaction or the Acquisition. This Agreement has been, or, as to Registration Rights Agreement to be executed after the date hereof, will be, duly executed and delivered by the Company and (assuming due authorization, execution and delivery by each other party thereto) each constitutes, a valid and binding obligation of the Company, enforceable against the Company in accordance with its terms, except as enforceability

may be limited by bankruptcy, insolvency, reorganization, moratorium and other similar laws relating to or affecting creditors' rights generally or by general equitable principles (regardless of whether such enforceability is considered in a proceeding in equity or at law).

- (b) <u>Preemptive Rights; Rights of First Offer.</u> None of the sale and issuance of the Purchased Shares pursuant to this Agreement and the issuance of Underlying Shares upon conversion of the Purchased Preferred Shares in accordance with their terms is or will be subject to any preemptive rights, rights of first offer or similar rights of any person (except as set forth in the Certificate of Designations or the Amended and Restated Charter).
- No Conflict. The Company is not in violation or default in any material respect of any provision of its Charter or Bylaws. The execution, delivery and performance by the Company of this Agreement and the Ancillary Agreements do not. and the consummation of the Transaction and compliance with the provisions hereof and thereof will not, result in a "change of control" (or similar event) under, or conflict with, or result in any default under, or give rise to an increase in, or right of termination, cancellation, acceleration or mandatory prepayment of, any obligation or to the loss of a benefit under, or result in the suspension, revocation, impairment, forfeiture or amendment of any term or provision of or the creation of any Encumbrance upon any of the properties or assets of the Company or any of its subsidiaries under, or require any consent or waiver under, any provision of (i) the Charter, the Bylaws or the comparable organizational documents of any of the Company's subsidiaries (other than the requirement to obtain stockholder approval as contemplated by Section 5.1), (ii) save in relation to the execution of the Credit Agreement, any Material Contract, or (iii) any applicable law, Judgment or Governmental Authorization, in each case applicable to the Company and its subsidiaries or their respective assets. Neither the execution and delivery of this Agreement, nor the consummation of the Transaction or the Acquisition, either alone or in combination with another event (whether contingent or otherwise) will (1) result in any payment of any "excess parachute payment" under Section 280G of the Code, (2) entitle any current or former employee, consultant or director of the Company or any of its subsidiaries to any payment, (3) increase the amount of compensation or benefits due to any such employee, consultant or director, or (4) accelerate the vesting, funding or time of payment of any compensation, equity award or other benefit.
- Section 5.7, no Governmental Authorization. Except as contemplated by Section 5.7, no Governmental Authorization, order or authorization of, or registration, qualification, declaration or filing with, or notice to, any Governmental Entity is required to be obtained or made by or with respect to the Company or any of its subsidiaries in connection with the execution, delivery and performance of this Agreement or any of the Ancillary Agreements or the Transaction, including the issuance of the Purchased Shares or the Underlying Shares (and any redemptions or conversions of the Purchased Preferred Shares pursuant to the terms thereof), except for such Governmental Authorizations, orders, authorizations, registrations, qualifications, declarations, filings and notices, the failure of which to be obtained or made would not (i) materially impair the Company's and/or Investor's ability to perform its obligations under this Agreement or the Ancillary Agreements or consummate the Transaction or (ii) have a Material Adverse Effect.

3.3 Capital Structure.

- As of the date of this Agreement, the authorized capital stock (a) of the Company consists solely of (i) 200,000,000 shares of Common Stock, of which 43,590,915 shares are issued and outstanding and 6,500,000 shares are reserved for issuance pursuant to the Company's 2008 Omnibus Incentive Plan, and (ii) 10,000,000 shares of preferred stock, of which zero (0) shares are issued and outstanding and zero (0) shares are reserved for issuance. As of the Closing, and after giving effect to the Transaction, the authorized capital stock of the Company shall consist solely of 400,000,000 shares of Common Stock and up to 20,000,000 shares of preferred stock, and the number of shares issued, outstanding and reserved for issuance shall consist solely of (x) the shares of Common Stock and preferred stock referred to in the foregoing sentence, (y) the Purchased Shares, and (z) a number of shares of Common Stock to be issued to the Designated Stockholders and to Markel Corporation in connection with the Designated Stockholder Investment at the Common Share Purchase Price and (xx) shares of Common Stock issued or reserved for issuance under the Company's 2008 Omnibus Incentive Plan, as outstanding in the SEC Report for the three months ended June 30, 2011 and any awards under the plan granted in the ordinary course of business since such SEC Report. No other shares of capital stock of the company are authorized, issued and outstanding or reserved for issuance.
- (b) Subject to the Company obtaining stockholder approval as contemplated by Section 5.1, the Purchased Shares have been duly authorized and upon the Closing shall be, and the Underlying Shares have been duly authorized and upon issuance shall be, (i) validly issued, fully paid and nonassessable, (ii) not issued in violation of any purchase option, call option, right of first refusal, preemptive right, subscription right or any similar right under any provision of the Charter or Bylaws of the Company or any Contract to which the Company or any of its subsidiaries is a party or by which any of its or their respective assets are bound, and (iii) free and clear of all Encumbrances.
- The Company has not issued any Voting Debt that is outstanding. Except as set forth in the SEC Reports or in Schedule 3.3(c) of the Company Disclosure Schedules, there are no (A) outstanding obligations, options, warrants, convertible securities, exchangeable securities, securities or rights that are linked to the value of the Common Stock or other rights, agreements or commitments relating to the capital stock of the Company or that obligate the Company to issue or sell or otherwise transfer shares of capital stock of the Company or any securities convertible into or exchangeable for any shares of capital stock of the Company or any Voting Debt of the Company, (B) outstanding obligations of the Company to repurchase, redeem or otherwise acquire shares of capital stock of the Company, (C) voting trusts, stockholder agreements, proxies or other agreements or understandings in effect with respect to the voting or transfer of shares of capital stock of the Company (but only to the Company's knowledge with respect to any such agreements to which neither the Company nor any subsidiary of the Company is a party), or (D) rights of first refusal, preemptive rights, subscription rights or any similar rights with respect to the capital stock of the Company under the Charter or Bylaws or any Contract to which the Company or any subsidiary of the Company is a party or by which any of its assets are bound. True, correct, and complete copies of the Charter and the Bylaws have been filed as exhibits to the Company's SEC Reports. Neither the

Company, nor any of its subsidiaries has any "stockholder rights plan", "poison pill" or any similar arrangement in effect.

The shares of outstanding capital stock of the Company's subsidiaries are duly authorized, validly issued, fully paid and nonassessable, and are held of record and beneficially owned by the Company or a subsidiary of the Company. There is no Voting Debt of any subsidiary of the Company that is outstanding. Except as set forth in the SEC Reports or in Schedule 3.3(d) of the Company Disclosure Schedules, there are no (i) outstanding obligations, options, warrants, convertible securities, exchangeable securities, securities or rights that are linked to the value of the Common Stock or other rights, agreements or commitments, in each case, relating to the capital stock or equity interests of the subsidiaries of the Company or that obligate the Company or its subsidiaries to issue or sell or otherwise transfer shares of the capital stock or any securities convertible into or exchangeable for any shares of capital stock or any Voting Debt of any subsidiary of the Company, (ii) outstanding obligations of the subsidiaries of the Company to repurchase, redeem or otherwise acquire shares of their respective capital stock or equity interests, (iii) voting trusts, stockholder agreements, proxies or other agreements or understandings in effect with respect to the voting or transfer of shares of capital stock of the subsidiaries of the Company (but only to the Company's knowledge with respect to any such agreements to which neither the Company nor any subsidiary of the Company is a party), or (iv) rights of first refusal, preemptive rights, subscription rights or any similar rights under any provision of the governing documents of any material subsidiary or any non-wholly owned subsidiary of the Company.

3.4 Taxes.

(a) Except as disclosed in the SEC Reports or except as would not reasonably be expected, individually or in the aggregate, to be material to the Company: (i) the Company and each of its subsidiaries have filed all Tax Returns required to be filed by them; (ii) all such Tax Returns are true, correct and complete; (iii) all Taxes due and owed by the Company and each of its subsidiaries (whether or not shown on any Tax Return) have been paid; (iv) neither the Company nor any of its subsidiaries currently is the beneficiary of any extension of time within which to file any material Tax Return; (v) no claim has ever been made by a Governmental Entity in a jurisdiction where the Company or any of its subsidiaries does not file Tax Returns that the Company or any such subsidiary is or may be subject to taxation by that jurisdiction; (vi) there are no liens on any of the assets or properties of the Company or any of its subsidiaries that arose in connection with any failure (or alleged failure) to pay any Tax; (vii) there are no ongoing, pending or, to the Company's knowledge, threatened audits, assessments, or other proceedings for or relating to any liability in respect of Taxes of the Company or any of its subsidiaries; (viii) neither the Company nor any of its subsidiaries has waived any statute of limitations in respect of Taxes or agreed to any extension of time with respect to a Tax assessment or deficiency; and (ix) the Company and each of its subsidiaries have timely withheld and paid all Taxes required to have been withheld and paid in connection with amounts paid or owing to any employee, independent contractor, creditor, stockholder or other third party.

(b) Neither the Company nor any of its subsidiaries has been a member of any affiliated group filing a consolidated Federal income Tax Return other than a group the common parent of which is the Company.

3.5 Compliance with Laws.

- (a) The Company and each of its subsidiaries and the conduct and operation of their respective businesses are and have been, since January 1, 2010, in compliance in all material respects with each applicable law that is applicable to the Company or its subsidiaries or their respective businesses, including any laws relating to employment practices, except, in each case, as has not had and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.
- (b) Neither the Company nor any of its subsidiaries has received any written notice or other written communication from any Governmental Entity or any other person regarding (1) any actual, alleged, possible or potential violation of or failure to comply with any material term or requirement of any Governmental Authorization, or (2) any actual, proposed, possible or potential revocation, withdrawal, suspension, cancellation, termination of or modification to any Governmental Authorization, except, in each case, as has not had and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.

3.6 SEC Reports and Company Financial Statements.

- (a) Since January 1, 2009, except as disclosed in Schedule 3.6(a) of the Company Disclosure Schedules, the Company has filed all reports, schedules, forms, statements and other documents required to be filed by it with the SEC pursuant to the reporting requirements of the Exchange Act (all the foregoing and all exhibits included or incorporated by reference therein and financial statements and schedules thereto and documents included or incorporated by reference therein being sometimes hereinafter collectively referred to as the "SEC Reports"). As of their respective filing dates, the SEC Reports complied with the requirements of the Exchange Act applicable to the SEC Reports (as amended or supplemented), and none of the SEC Reports, at the time they were filed with the SEC, contained any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. No subsidiary of the Company is, or is required to be, a registrant with the SEC.
- (b) As of their respective dates, except as set forth therein or in the notes thereto, the financial statements contained in the SEC Reports and the related notes (the "Financial Statements") complied as to form with all applicable accounting requirements and the published rules and regulations of the SEC with respect thereto. The Financial Statements (i) were prepared in accordance with GAAP, consistently applied during the periods involved (except (x) as may be otherwise indicated in the notes thereto or (y) in the case of unaudited interim statements, to the extent that they may not include footnotes or may be condensed or summary statements), (ii) fairly present in all material respects the consolidated financial position of the Company and its consolidated

subsidiaries as of the dates thereof and the consolidated results of their operations and cash flows for the periods then ended (subject, in the case of unaudited statements, to normal year-end audit adjustments), and (iii) are in all material respects in accordance with the books of account and records of the Company and its consolidated subsidiaries.

- (c) Neither the Company nor any of its subsidiaries is a party to, or has any commitment to become a party to, any joint venture, off-balance sheet partnership or any similar Contract or arrangement (including any Contract relating to any transaction or relationship between or among the Company and any of its subsidiaries, on the one hand, and any unconsolidated affiliate of the Company or any of its subsidiaries, including any structured finance, special purpose or limited purpose entity or person, on the other hand, or any "off-balance sheet arrangements" (as defined in Item 303(a) of Regulation S-K of the SEC)), where the result, purpose or effect of such Contract is to avoid disclosure of any material transaction involving, or material liabilities of, the Company or any of its subsidiaries in the Company's or such subsidiary's audited financial statements or other SEC Reports.
- (d) To the Company's knowledge, there is no reason that its outside auditors and its chief executive officer and chief financial officer will not be able to give the attestations and certifications, as applicable, required pursuant to the rules and regulations adopted pursuant to Section 404 of SOX, without qualification, when next due.
- (e) Except as set forth in SEC Reports and the transactions contemplated by the Designated Stockholder Investment, none of the executive officers, directors or related persons (as defined in Regulation S-K Item 404) of the Company is presently a party to any transaction with the Company or any of its subsidiaries that would be required to be reported on Form 10-K by Item 13 thereof pursuant to Regulation S-K Item 404.
- 3.7 Private Placement. Assuming that the representations of the Investor set forth in Section 4.3 are true and correct, the offer, sale, and issuance of the Purchased Shares and the issuance of the Underlying Shares upon conversion of the Purchased Preferred Shares, in each case in conformity with the terms of this Agreement, are exempt from the registration requirements of Section 5 of the Securities Act.
- 3.8 Brokers. Except for fees payable to BDT & Company, LLC, the Company and its subsidiaries have incurred no obligation or liability, contingent or otherwise, in connection with this Agreement or the Transaction that would result in the obligation of the Investor to pay any finder's fees, brokerage or agent's commissions or other like payments in connection with the negotiations leading to this Agreement or the consummation of the Transaction.
- 3.9 Registration Rights. Other than the Registration Rights Agreement, and the registration rights agreements to be entered into with the Designated Stockholders and Markel Corporation in connection with the Designated Stockholder Investment, true, correct, and complete forms of which have been provided to the Investor, the Company has not granted or agreed to grant, and is not under any obligation to provide, any rights

(including "piggy-back" registration rights) to register under the Securities Act any of its presently outstanding securities or any of its securities that may be issued subsequently.

- 3.10 No Restriction on the Ability to Pay Cash Dividends. Except as set forth in the Amended and Restated Charter, the Certificate of Designations and the Credit Agreement and any existing credit facilities of the Company or its subsidiaries that are being replaced by the Credit Agreement, neither the Company nor any of its subsidiaries is, or will be, immediately following the Closing, a party to any Contract, and is not, and will not be, immediately following the Closing, subject to any provisions in its Charter or Bylaws or other governing documents or resolutions of the Board of Directors or other governing body, that restricts, limits, prohibits or prevents the payment of cash dividends with respect to any of its equity securities.
- 3.11 Credit Agreement. True, correct and complete copies of the Credit Agreement and all ancillary agreements thereto that have been executed as of the date hereof have been provided to the Investor, and the Credit Agreement and each such ancillary agreement is in full force and effect.

ARTICLE 4

Representations and Warranties of the Investor

The Investor hereby represents and warrants to the Company, as of the date hereof and as of the Closing Date, as follows:

4.1 Entity Status. The Investor is duly incorporated or otherwise organized, validly existing and in good standing under the applicable laws of its governing jurisdiction and has all requisite corporate or other power and authority to carry on its business as it is now being conducted, and no proceeding has been instituted in any such jurisdiction revoking, limiting or curtailing, or seeking to revoke, limit or curtail, such power, authority or qualification.

4.2 Authorization; Noncontravention.

and authority to execute and deliver this Agreement and the Ancillary Agreements to which it is a party, to perform its obligations hereunder and thereunder and to consummate the Transaction. The execution, delivery and performance of this Agreement and the Ancillary Agreements to which it is a party and the consummation by the Investor of the Transaction have been duly and validly authorized by all necessary entity action, and no other entity proceedings on the part of the Investor or vote of holders of any class or series of capital stock or equity interests of the Investor is necessary to authorize this Agreement and the Ancillary Agreements to which the Investor is a party or to consummate the Transaction. This Agreement and the Registration Rights Agreement to which the Investor is a party have been duly executed and delivered by the Investor and (assuming due authorization, execution and delivery by each other party thereto) each constitutes a valid and binding obligation of the Investor, enforceable against the Investor in accordance with its respective terms, except as enforceability may be limited by bankruptcy, insolvency,

reorganization, moratorium and other similar laws relating to or affecting creditors' rights generally or by general equitable principles (regardless of whether such enforceability is considered in a proceeding in equity or at law).

- No Conflict. The Investor is not in violation or default of any provision of its organizational documents. The execution, delivery and performance by the Investor of this Agreement and the Ancillary Agreements to which it is a party do not, and the consummation of the Transaction and compliance with the provisions of this Agreement and the Ancillary Agreements to which it is a party will not, conflict with, or result in any default under, any provision of (i) the organizational documents of the Investor, (ii) any material Contract to which the Investor is a party or by which any of the Investor's assets are bound, or (iii) any applicable law, Governmental Authorization or Judgment, in each case applicable to the Investor, other than, in the case of clauses (ii) and (iii), any such conflicts or defaults that would not reasonably be expected to materially impair or delay the ability of the Investor to perform its obligations under this Agreement or the Ancillary Agreements to which it is a party or carry out the Transaction in accordance with the terms hereof or thereof. Except as contemplated by Section 5.7, no Governmental Authorization, order or authorization of, or registration, qualification, declaration or filing with, or notice to, any Governmental Entity is required to be obtained or made by or with respect to the Investor in connection with the execution, delivery and performance of this Agreement or any of the Ancillary Agreements to which the Investor is a party or the other Transaction, except for such Governmental Authorizations, orders, authorizations, registrations, declarations, filings and notices, the failure of which to be obtained or made would not materially impair the Investor's ability to perform its obligations under this Agreement or the Ancillary Agreements or consummate the Transaction.
- 4.3 Securities Act; Purchase for Investment Purposes. The Investor (i) is acquiring the Purchased Shares solely for investment with no present intention to distribute them in violation of the Securities Act and the rules and regulations thereunder or any applicable U.S. state securities laws, (ii) has such knowledge and experience in financial and business matters and in investments of this type that it is capable of evaluating the merits and risks of making an informed investment decision to purchase the Purchased Shares, (iii) is an "accredited investor" (as that term is defined by Rule 501 promulgated under the Securities Act), and (iv) acknowledges and understands that the Purchased Shares have not been registered under the Securities Act, or any state securities laws, and agrees that it will not offer, sell, assign, transfer, pledge, encumber or otherwise dispose of such Purchased Shares (including the Underlying Shares) absent registration under the Securities Act or unless such transaction is exempt from, or not subject to, registration under the Securities Act, and in each case, in accordance with all applicable state securities laws. The Investor did not learn of the opportunity to purchase the Purchased Shares by means of any form of general or public solicitation or general advertising, or publicly disseminated advertisements or sales literature, including (a) any advertisement, article, notice or other communication published in any newspaper, magazine, or similar media, or broadcast over television or radio, or (b) any seminar or meeting to which the Investor was invited by any of the foregoing means of communications. The Investor understands that the Purchased Shares will be subject to substantial contractual and legal restrictions on transferability. The Investor understands that the Purchased Shares will be characterized

as "restricted securities" under the United States federal securities laws inasmuch as they are being acquired from the Company in a transaction not involving a public offering and that under such laws and applicable regulations such securities may be resold without registration under the Securities Act only in certain limited circumstances. The Investor understands that no United States federal or state agency or any other government or governmental agency has passed on or made any recommendation or endorsement of the Purchased Shares or the fairness or suitability of the Purchased Shares. The Investor understands that until such time as the resale thereof has been registered under the Securities Act, certificates evidencing the Purchased Shares shall bear the following legend:

THE SECURITIES REPRESENTED BY THIS CERTIFICATE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "SECURITIES ACT"), OR ANY STATE SECURITIES LAWS. NEITHER THE SECURITIES REPRESENTED BY THIS CERTIFICATE NOR ANY INTEREST OR PARTICIPATION HEREIN MAY BE OFFERED, SOLD, ASSIGNED, TRANSFERRED, PLEDGED, ENCUMBERED, OR OTHERWISE DISPOSED OF IN ABSENCE OF SUCH REGISTRATION OR UNLESS SUCH TRANSACTION IS EXEMPT FROM, OR NOT SUBJECT TO. REGISTRATION UNDER THE SECURITIES ACT, AND, IN EACH CASE, IN ACCORDANCE WITH ALL APPLICABLE SECURITIES LAWS OF THE STATES OF THE UNITED STATES OR UPON DELIVERY TO THE COMPANY OF AN OPINION OF COUNSEL AN EXEMPTION FROM SUCH REGISTRATION OR QUALIFICATION IS APPLICABLE.

The Investor will obtain representations from the Co-Investors similar to those in Section 4.3 with respect to their direct or indirect investments in the Investor, and the Investor shall make copies of such representations available to the Company.

- 4.4 Brokers. The Investor has incurred no obligation or liability, contingent or otherwise, in connection with this Agreement that would result in the obligation of the Company or any of its Affiliates to pay any finder's fees, brokerage or agent's commissions or other like payments in connection with the negotiations leading to this Agreement or the consummation of the Transaction.
- 4.5 Commitment Letters. True, correct and complete copies of the Commitment Letters have been provided to the Company.
- 4.6 Code Matters. Neither the Investor nor any Person that would be deemed to be acting in concert with the Company as a result of its connection with the Investor has acquired any interest in the securities of Target in the twelve (12) months preceding the date of this Agreement, nor will the Investor nor any such person acquire any such interest in securities of the Target or take any other action which would affect the terms on which any offer must be made to the Target prior to the earlier of (i) the closure of the Acquisition, and (ii) the termination of this Agreement. For purposes of this Section 4.6, "acting in concert" has the meaning given in the City Code.

ARTICLE 5

Covenants

- 5.1 Stockholder Approvals. The Company shall present and recommend to the stockholders of the Company at a special meeting of the stockholders of the Company (to be held in accordance with the corporate laws of Delaware, the Charter and the Bylaws), a proposal to approve the Transaction and the Designated Stockholder Investment and to approve and adopt the Amended and Restated Charter. The Company will provide the Investor and its counsel, in writing, copies of any draft proxy statement to be distributed to the Company's shareholders in connection with such meeting, prior to the distribution thereof, as well as any comments or other communications, whether written or oral, that the Company or its counsel may receive from time to time from the SEC or its staff with respect to such proxy statement promptly after the receipt of such comments or other communications, and will provide the Investor and its counsel with the opportunity to review and comment on such proxy statement and the Company's responses to any such comments or communications received from the SEC or its staff.
- other and cooperate with each other prior to any public disclosure of the Transaction. The Company, on the one hand, and the Investor, on the other hand, agree that no public release or announcement concerning the Transaction shall be issued by or as a result of the actions of any of them without the prior consent of the other, which consent shall not be unreasonably withheld or delayed, except in accordance with Section 5.6 or as such release or announcement may be required by applicable law or the rules and regulations of the New York Stock Exchange, the Financial Services Authority, the Takeover Panel or the City Code, in which case the party required to make the release or announcement shall use its reasonable best efforts to consult with the other parties hereto about, and allow the other parties hereto to reasonably comment on, such release or announcement in advance of such issuance.

5.3 Certain Tax Matters.

- (a) The Company and the Investor acknowledge and agree that the Purchased Shares are being issued solely in consideration of the Investment Amount and neither the Company nor the Investor shall take any position for financial accounting, Tax or other purposes inconsistent with such agreement. Without limiting the foregoing, the Company shall not record an expense or apply any withholding (other than any withholding that may be required by law pursuant to Section 1441, 1442, 1445 or other applicable provision of the Code and comparable state or local laws) in connection with the issuance or any exercise or redemption of, any adjustment to, or any payments made in respect of any of the Purchased Shares.
- (b) The Company and the Investor acknowledge and agree that the exercise of the Company's right to optionally redeem the Purchased Preferred Shares pursuant to the Certificate of Designations is not more likely than not to occur within the meaning of Treasury Regulation Section 1.305 5(b)(3).

5.4 Standstill Agreement.

- (a) The Investor agree that, without the prior approval of the Company, the Investor will not, directly or indirectly, or in concert with any person, or as participants in a "group" (as defined in Section 13 of the Exchange Act):
- (i) Make any public announcement with respect to, or submit to the Company or any of its directors, officers, representatives, trustees, employees, attorneys, advisors, agents or Affiliates, any proposal from the Investor or its Affiliates for, the acquisition of any Voting Stock of the Company or with respect to any merger, consolidation, business combination or purchase of any substantial portion of the assets of the Company, whether or not such proposal might require the making of a public announcement by the Company unless the Company shall have made a prior written request to the Investor to submit such a proposal;
- (ii) Except for (A) actions taken by directors of the Company who are affiliated with the Investor in their capacities as directors and (B) actions taken for the purpose of electing to the Company's Board of Directors the nominees of the Investor in accordance with the Amended and Restated Charter, the Investor will not (1) call or seek to call a special meeting of the stockholders of the Company or of any of the subsidiaries of the Company for action by stockholders of the Company or of any of the Company subsidiaries, (2) make, or in any way participate in, any "solicitation" of "proxies" (as such terms are defined or used in Regulation 14A under the Exchange Act) to vote any Voting Stock of the Company, or (3) become a "participant" in any "election contest" as such terms are defined or used in Rule 14a-11 under the Exchange Act with respect to any voting securities of the Company; or
- (iii) Effect or seek any recapitalization, reclassification, liquidation or dissolution or other extraordinary transaction that would have the effect of any of the matters described in Sections 5.4(a)(i) or 5.4(a)(ii).
- (b) The Investor's obligations under this Section 5.4 shall terminate and be of no further force and effect on the first date that the Investor either (1) has beneficial ownership (as defined in Rule 13d-1 promulgated under the Exchange Act) of less than 5% of the outstanding Common Stock (including securities convertible into Common Stock), or (2) no longer has the right to nominate for election as least one member of the Board of Directors as set forth in the Amended and Restated Charter.
- 5.5 Credit Agreement. Prior to the Closing Date, the Company shall not, and shall cause its subsidiaries not to, amend or waive any material term or condition of the Credit Agreement, the effect of which amendment or waiver would be to alter the principal amount, covenants, interest, or fees payable under the Credit Agreement in a manner materially adverse to the Company or the Investor, without the prior written consent of the Investor; it being understood that this shall not restrict the Company from (i) exercising any of the "Incremental Facilities" provisions contained in Section 2.14 of the Credit Agreement in accordance with the terms of such provisions existing on the date hereof, or (ii) making any amendments necessary to reflect the exercise of any "flex" rights that have been agreed with the arrangers under the Credit Agreement or any amendments to the

Credit Facility as a result of a change in the amount of debt funding as contemplated by the second sentence of Section 5.13. The Company shall use its best efforts to cause all conditions precedent to the effectiveness of the Credit Agreement and to the extensions of credit for the purposes of Certain Funds Credit Extension (as defined in the Credit Agreement) only to be satisfied on or before the Closing Date.

- 5.6 Acquisition Covenants. The Company shall, and shall cause its subsidiaries to perform and comply with each of the following covenants in connection with the Acquisition:
- (a) The Company and its subsidiaries will not amend or waive any material term or condition of the Offer Document or, as the case may be, Scheme Circular without the prior written consent of the Investor, except for any such amendment or waiver as (i) does not materially prejudice the Investor, (ii) is required by the Takeover Panel, the High Court of Jersey or any other applicable law, regulation or regulatory body, or (iii) extends the period in which holders of Target Shares may accept the terms of the Offer or the Scheme, as the case may be.
- (b) Subject to <u>Section 5.13</u>, if the Acquisition is implemented by way of an Offer, the Company and it subsidiaries will not reduce the acceptance condition to below the level required to allow the Company to give notice under Section 117 of the Companies Act in respect of the Target Shares.
- (c) The Company and its subsidiaries shall comply in all respects with the City Code (subject to any waiver granted by the Takeover Panel) and all other applicable laws (including the Financial Services and Markets Act 2000 (as amended)) and/or regulations relating to any Scheme of Arrangement or, as the case may be, Offer, except to the extent that such non-compliance would not be materially prejudicial to the Investor.
- (d) The Company and its subsidiaries shall not, without the prior written consent of the Investor, increase the price to be paid for any Target Shares pursuant to a Scheme of Arrangement or, as the case may be, an Offer.
- (e) The Company and its subsidiaries shall not take any action that would require the Company or any of its subsidiaries to make a mandatory offer for the Target Shares in accordance with Rule 9 of the City Code.
- (f) In the event the Acquisition is made pursuant to an Offer, where becoming permitted to do so, the Company shall promptly give notices under Article 117 of the Companies Act in respect of the Target Shares.
- (g) The Company and its subsidiaries shall, upon reasonable request and in each case to the extent that they are able to do so in compliance with applicable law and confidentiality or other obligations to which they are subject as of the date of this Agreement, keep the Investor informed as to the status and progress of (or otherwise relating to) an Offer (and, in the case of an Offer, the current level of acceptances in respect of that Offer) or, as the case may be, a Scheme of Arrangement;

- (h) The Company and its subsidiaries shall, to the extent that they are able to do so in compliance with applicable law and confidentiality or other obligations to which they are subject as of the date of this Agreement, promptly supply to the Investor (i) copies of all documents, certificates, notices or announcements received or issued by the Company or any of its subsidiaries (or on their behalf) in relation to an Offer or a Scheme of Arrangement (as the case may be) and (ii) any other information regarding the progress of an Offer or a Scheme of Arrangement (as the case may be) in each case as the Investor may reasonably request.
- (i) Other than as required by the Takeover Panel, the City Code, the London Stock Exchange (including any stock exchange of which it is the operator), the Financial Services Authority, the Exchange Act (and the rules and regulations thereunder) or any other applicable law, regulation or regulatory body, the Company and its subsidiaries shall not make any press release or other public statement in respect of the Acquisition (other than in the Rule 2.5 Announcement, any Offer Document or any Scheme Circular), without first obtaining the prior approval of the Investor, such approval not to be unreasonably withheld or delayed.
- (j) The Company and its subsidiaries shall procure that as soon as reasonably practicable, and in any event within twenty (20) Business Days, after the earlier of: (i) an office copy of the Scheme Court Order and the original and an office copy of the Reduction Court Order, attaching a minute (within the meaning of Article 64(1) of the Companies Act) of such reduction, having been delivered to the Registrar of Companies for registration, and, in relation to the reduction of ordinary share capital, the Reduction Court Order having been registered, and (ii) Parent having become, directly or indirectly, the legal and beneficial owner of at least 75% of the voting shares in the Target, the Target is delisted and re-registered as a private company.
- (k) The Company and its subsidiaries shall not issue any Offer Document or, as the case may be, approve the issuance of any Scheme Circular, that makes reference to Investor or a Co-Investor or any of their respective Affiliates, in each case, without the final proof of the same being approved in writing by the Investor, such approval not to be unreasonably withheld or delayed, *provided*, that if such approval is unreasonably withheld or delayed by the Investor, the Company or its subsidiary, as the case may be, shall be entitled to proceed forthwith in issuing any Offer Document or Scheme Circular as required by the Takeover Panel, the City Code or any other applicable regulatory body or law or regulation.
- (l) The Company and its subsidiaries shall not, without the prior written consent of the Investor, enter into any material agreement or understanding of any nature (whether formal or informal) with the trustees of any pension scheme of the Target that would result in an incremental annual cash expense of greater than \$5,000,000 or a settlement payment at Closing of greater than \$50,000,000, such consent not to be unreasonably withheld or delayed.
- (m) The Company and its subsidiaries shall not, without the prior written consent of the Investor, enter into any material agreement or understanding of any nature (whether formal or informal) with any anti-trust or other regulatory authority

relating to the Acquisition to the extent that any such agreement or understanding relates to a disposition, partial disposition, restructuring, reorganization, hold separate or similar transaction involving a line of business of the Company or its subsidiaries accounting for \$100,000,000 or more of gross revenues during any of the last five calendar years, such consent not to be unreasonably withheld or delayed.

- (n) Save in relation to any condition as to acceptances if the method of effecting the Acquisition changes from a Scheme of Arrangement to an Offer, the Company and its subsidiaries shall (subject to obtaining the consent of the Takeover Panel in accordance with the City Code) invoke any condition to the Acquisition which becomes capable of being invoked unless the Investor consents in writing to such condition not being invoked.
- (o) The Company and its subsidiaries shall not, for the purposes of Note 1 to Rule 21.1 of the Takeover Code, consent to the taking of any action by the Target (or any member of its group) which would otherwise amount to frustrating action without the prior written consent of the Investor.
- (p) The Company and its subsidiaries shall use the proceeds of the sale of the Purchased Shares solely for the purpose of the Acquisition and for no other purpose. The Company and its subsidiaries shall use the proceeds of all borrowings under the Credit Agreement solely for purposes of the Acquisition and the other purposes permitted under the Credit Agreement.

5.7 HSR Act Filings; Competition Filings.

(a) Filings and Cooperation. Each of the Company and the Investor will take all reasonable steps to: (i) as soon as reasonably practicable following the date of this Agreement, make or cause to be made the filings required of such party or any of its Affiliates or subsidiaries under the HSR Act with respect to the Transaction; (ii) except as otherwise provided in the following sentence, as soon as reasonably practicable following the date of this Agreement, make or cause to be made any other filings with all foreign Governmental Entities in any jurisdiction in which the parties believe it is necessary or advisable; (iii) comply in a timely manner with any request under any antitrust or competition laws of any jurisdiction ("Antitrust Laws") for additional information, documents, or other material received by such party or any of its affiliates or subsidiaries from the Federal Trade Commission or the Department of Justice or other Governmental Entity in respect of such filings or the Transaction; and (iv) cooperate with the other party in connection with any such filings and in connection with resolving any investigation or other inquiry of any such agency or other Governmental Entity under any Antitrust Laws with respect to any such filing or the Transaction. With regard to any communication with any Governmental Entity regarding such filings, each party will inform the other party: (1) before delivering any material communication to a Governmental Entity, (2) promptly after receiving any material communication from a Governmental Entity, and (3) before entering into any proposed understanding, undertaking, or agreement with any Governmental Entity regarding any such filings or the Transaction. Neither party will participate in any meeting with any Governmental Entity in respect of any such filings, investigation, or other inquiry without giving the other party prior notice of the meeting and, to the extent permitted by such Governmental Entity, the opportunity to attend and participate to the extent permitted by applicable law and to the extent reasonably practicable under the circumstances. Each of the Company and the Investor will take such reasonable action as may be required to cause the expiration of the notice periods under the HSR Act or similar laws or requirements of any Governmental Entity with respect to the Transaction as promptly as possible after the execution of this Agreement

(b) <u>Filing Fees.</u> The Company shall pay one hundred percent (100%) of the filing fees applicable under the HSR Act with respect to the Transaction.

5.8 Transfer Restrictions.

- (a) The Investor shall not sell or otherwise transfer (a) any of the Purchased Common Shares to any person during the first 180 days following the Closing Date, or (b) greater than fifty percent (50%) of the Purchased Common Shares to any person between the 180th day following the Closing Date and the first anniversary of the Closing Date. The period from the Closing Date through and including the second anniversary of the Closing Date is referred to in this Agreement as the "Common Share Lock Up Period". Upon the conversion of any Purchased Preferred Shares into shares of Common Stock, the Investor shall not sell or otherwise transfer any such shares of Common Stock to any person during the first 180 days following the date of such conversion or (y) greater than fifty percent (50%) of such shares to any person between the 180th day following the date of such conversion and the first anniversary of the date of such conversion.
- (b) The Investor shall not sell or otherwise transfer any shares of Common Stock, Purchased Preferred Shares or Underlying Shares to any Third Party that, to the knowledge of the Investor, (a) is a Competitor of the Company, or (b) is, or would as a result of such sale or transfer become, the beneficial owner (as defined in Section 13 of the Exchange Act) of five percent (5%) or more of the outstanding capital stock of the Company.
- (c) The provisions of Section 5.8(b) (and in the case of subclauses (i), (ii), (v) and (vi) below, the provisions of Section 5.8(a)) shall not apply to any sale or other transfer that is (i) pursuant to a tender or exchange offer for a majority of the outstanding shares of the Company; (ii) pursuant to a merger or consolidation of the Company; (iii) pursuant to a widely distributed underwritten public offering, including pursuant to the Registration Rights Agreement; (iv) pursuant to Rule 144 under the Securities Act; (v) a transfer or distribution to any Permitted Transferee (provided that such Permitted Transferee agrees pursuant to a written instrument that is reasonably acceptable to the Company to comply with all of the restrictions on transfer that are applicable to the Investor); or (vi) in connection with a Tag-Along Transaction (as defined in Section 5.12).
- 5.9 Information Rights. Until the earlier of (a) the end of the Common Share Lock Up Period, or (b) the date that the Investor owns less than twenty percent (20%) of aggregate shares of Common Stock on an as-converted, fully diluted basis (treating the

Purchased Preferred Shares as fully converted into shares of Common Stock but excluding from this calculation any shares of Common Stock issuable upon the exercise of other Common Share Equivalents or upon the exercise of options or restricted stock units issued to employees, consultants or directors of the Company pursuant to any equity incentive plan), the Company shall provide the Investor with (i) monthly unaudited financial information, including monthly balance sheets and income statements, no later than thirty (30) days following the end of each month, and (ii) such other financial and operational information as the Investor may reasonably request. The Investor shall maintain the confidentiality all information provided pursuant to this Section 5.9 in accordance with Regulation FD promulgated under the Exchange Act. The Investor shall be permitted to share all information provided pursuant to this Section 5.9 with the Co-Investors, provided, that the Co-Investors shall be bound to maintain the confidentiality of such information in accordance with Regulation FD on terms reasonably satisfactory to the Company.

- 5.10 Drawdown of Equity Commitments. Subject to satisfaction of the conditions set forth in Sections 6.1 and 6.3 (other than those that by their nature are to satisfied by the taking of actions at the Closing), the Investor shall provide written notice to the Co-Investors no later than the date on which the Offer becomes unconditional in all respects, or if the Acquisition is implemented by way of a Scheme of Arrangement, the date on which the Scheme of Arrangement becomes effective in accordance with its terms that their respective capital contributions to the Investor are due and payable in accordance with the terms and provisions of the Commitment Letters, and, when such capital contributions are received by the Investor, the Investor shall use the proceeds thereof to pay the Investment Amount due and payable to the Company at the Closing pursuant to Section 2.2.
- 5.11 BDT Fund Capital Commitment. Subject to satisfaction of the conditions set forth in Sections 6.1 and 6.3 (other than those that by their nature are to be satisfied by the taking of actions at the Closing) and to a partial assignment pursuant to Section 8.2, the BDT Fund shall make a capital contribution to the Investor on or before the Closing Date in the amount of \$390,000,000 (such \$390,000,000, the "Fund Commitment Amount") (\$30,000,000 of which will be deemed to be satisfied by the capital contribution made by BDT Main to the Investor pursuant to the terms of the BDT Main Equity Commitment Letter), which the Investor shall use to pay the Investment Amount due and payable to the Company at the Closing pursuant to Section 2.2.

5.12 Tag-Along Rights.

(a) If either or both of the Designated Stockholders shall propose to sell or otherwise transfer any shares of Common Stock or Common Stock Equivalents (a "<u>Tag-Along Transaction</u>") to any person (other than to an immediate family member or a trust that such immediate family member or a Designated Stockholder controls, or as to which a Designated Stockholder or such immediate family member is the primary beneficiary; *provided*, that such immediate family member or such trust shall have agreed to be bound by the provisions of this <u>Section 5.12</u> on terms reasonably satisfactory to the Investor) (a "<u>Proposed Transferee</u>"), such proposed Tag-Along Transaction shall be conditioned upon satisfaction of the requirements set forth in this Section 5.12.

- (b) Not more than sixty (60) days and not less than thirty (30) days prior to the effective date of the proposed Tag-Along Transaction (the "Transfer Date"), each Designated Stockholder participating in a Tag-Along Transaction shall deliver to the Investor a binding written offer (a "Tag-Along Offer") (conditioned solely upon the consummation of such proposed sale) from such Designated Stockholder or the Proposed Transferee(s) to purchase, at the same price and upon terms and conditions identical in all respects to the terms and conditions on which such Designated Stockholder proposes to sell or transfer shares of Common Stock or Common Stock Equivalents to the Proposed Transferee, a percentage of the Investor's shares of Common Stock and/or Common Stock Equivalents (including any Underlying Shares) equal to the percentage of such Designated Stockholder's (or Designated Stockholders') shares of Common Stock and Common Stock Equivalents proposed to be included in the Tag-Along Transaction.
- The Tag-Along Offer shall specify, in reasonable detail, the number of shares of Common Stock and/or Common Stock Equivalents proposed to be included in the Tag-Along Transaction, the proposed purchase price, the Transfer Date, the identity of the Proposed Transferee(s), and the other terms and conditions of the proposed Tag-Along Sale. Upon receipt of a Tag-Along Offer, the Investor may elect to either (i) accept the Tag-Along Offer in whole or in part by delivering written notice of such acceptance not less than ten (10) Business Days prior to the Transfer Date specified in such * Tag-Along Offer, to the participating Designated Stockholder or to the Proposed Transferee(s), as applicable, which notice shall specify the number of shares of Common Stock and/or Common Stock Equivalents that the Investor intends to sell in the Tag-Along Transaction, or (ii) sell or otherwise transfer to a Third Party, on such terms as the Investor shall determine, at any time following delivery of a Tag-Along Offer, a percentage of the Investor's shares of Common Stock and Common Stock Equivalents equal to the percentage of shares of Common Stock and Common Stock Equivalents proposed to be sold or otherwise transferred by the Designated Stockholder in such Tag-Along Offer; provided, that any sale pursuant to clause (ii) shall comply with Section 5.8(b).
- (d) If the offer of the Proposed Transferee(s) or Designated Stockholders states that the Proposed Transferee(s) or the Designated Stockholders is or are unwilling to purchase, in the aggregate, more than a specified amount of Common Stock or Common Stock Equivalents, and the Investor elects to accept the Tag-Along Offer, the Common Stock and Common Stock Equivalents proposed to be sold by the Investor and the Designated Stockholder in the Tag-Along Sale shall be allocated *pro rata* in accordance with their relative holdings of shares of Common Stock and Common Stock Equivalents.
- 5.13 Acceptance Condition. Notwithstanding Section 5.6(b), if the Acquisition is implemented by way of an Offer, at the written request of the Company, the acceptance condition of the Offer may be reduced below the level required to allow the Company to give notice under Section 117 of the Companies Act in respect of the Target Shares, provided, however, that in no event shall the acceptance condition be reduced below seventy-five percent (75%) in nominal value of the shares to which the Offer relates. In the event of a reduction of the acceptance condition pursuant to this Section 5.13, the parties shall negotiate in good faith in order to amend this Agreement to the extent

reasonably necessary to appropriately reflect the revised structure of the Acquisition, including by adjusting the amount of equity and debt funding for the Company.

5.14 Code Matters.

- (a) Following the earlier of (i) the closure of the Acquisition, and (ii) the termination of this Agreement, the Investor will not directly or indirectly take any action that would result in a mandatory offer for securities of the Target pursuant to Rule 9 of the City Code being required, unless the Takeover Panel has expressly confirmed that such action has no concert party consequences under the City Code for the Company.
- (b) Nothing in this agreement shall prohibit or otherwise restrict dealings in securities of the Target by: (i) Persons which are not presumed to be Persons acting in concert with the Investor under Rule 7.2(c) of the City Code; or (ii) entities in respect of which the Takeover Panel has expressly confirmed that dealings in the securities of the Target have no concert party consequences under the City Code for either of the Investor or the Company.
- (c) For purposes of this <u>Section 5.14</u>, "acting in concert" has the meaning given in the City Code.

ARTICLE 6

Conditions

- 6.1 Conditions to all Parties' Obligations. The respective obligations of the Company, on the one hand, and the Investor, on the other hand, to effect the Closing are subject to the satisfaction of the following conditions:
- (a) <u>HSR Act.</u> Any applicable waiting periods, together with any extensions thereof, under the HSR Act shall have expired or been terminated.
- (b) <u>No Order.</u> No Governmental Entity of competent jurisdiction will have enacted, issued, promulgated, enforced, or entered any statute, rule, regulation, executive order, decree, injunction, or other order (whether temporary, preliminary, or permanent) that (a) is in effect, and (b) has the effect of making the Transaction illegal or otherwise prohibiting consummation of the Transaction.
- Charter and Certificate of Designations. All approvals of the Company's stockholders required in connection with the Transaction, including any such approvals required under applicable law and the applicable listing rules of the New York Stock Exchange, shall have been obtained on or before the Closing Date, and the Amended and Restated Charter and the Certificate of Designations, substantially in the forms attached hereto as **Exhibit A** and **Exhibit B**, respectively, shall have been properly filed with the office of the Secretary of State of the State of Delaware.
- (d) Offer/Scheme of Arrangement. In the case of an Offer, the Offer shall have become or been declared unconditional in all respects. In the case of a

Scheme of Arrangement, the High Court of Jersey shall have filed an order (and the Investor shall have received a copy thereof) sanctioning the Scheme of Arrangement on behalf of the Target with the Registrar of Companies in accordance with Part 18A of the Companies Act.

- 6.2 Conditions to the Company's Obligations. The obligations of the Company to effect the Closing are subject to the satisfaction, or written waiver by the Company (given in accordance with Section 8.1), of each of the following conditions:
- (a) Representations and Warranties. Each of the representations and warranties of the Investor contained in this Agreement and the Ancillary Documents that are not qualified as to materiality or material adverse effect (or any correlative term) shall be true and correct in all material respects on and as of the date of this Agreement and as of the Closing with the same force and effect as if made as of the Closing (except that representations and warranties which address matters only as of a particular date need only be true and correct as of such date), and each of the representations and warranties of the Investor contained in this Agreement and the Ancillary Documents that are qualified as to materiality or material adverse effect (or any correlative term) shall be true and correct in all respects on and as of the date of this Agreement and as of the Closing with the same force and effect as if made as of the Closing (except that representations and warranties which address matters only as of a particular date need only be true and correct as of such date).
- (b) <u>Covenants</u>. The covenants and agreements set forth in this Agreement and in the Ancillary Agreements to be performed or complied with by the Investor at or prior to the Closing shall have been performed or complied with in all material respects.
- 6.3 Conditions to the Investor's Obligations. The obligations of the Investor to effect the Closing are subject to the satisfaction, or written waiver by the Investor (given in accordance with Section 8.1), of each of the following conditions:
- (a) Representations and Warranties. Each of the Major Representations of the Company contained in this Agreement shall be true and correct in all material respects on and as of the date of this Agreement and as of the Closing with the same force and effect as if made as of the Closing (except that representations and warranties which address matters only as of a particular date need only be true and correct as of such date).
- (b) <u>Covenants.</u> The Major Undertakings set forth in this Agreement to be performed or complied with by the Company at or prior to the Closing shall have been performed or complied with in all material respects.
- (c) <u>Debt Financing.</u> As of the Closing Date, (i) the Credit Agreement shall be in full force and effect, and (ii) all conditions precedent to the effectiveness of the Credit Agreement and to the extensions of credit thereunder (other than receipt of the proceeds of the Equity Financing (as defined in the Credit Agreement) shall have been satisfied or waived.

ARTICLE 7

Termination

- 7.1 Termination. This Agreement may be terminated prior to the Closing as follows:
- (a) At any time, by the mutual written consent of the Investor and the Company;
- (b) By the Investor if the Company does not publish the Rule 2.5 Announcement by September 15, 2011 or, if the Rule 2.5 Announcement is published by such date, on such later date, if any, upon which the Scheme of Arrangement or Offer, as applicable, lapses, terminates or is withdrawn;
- (c) By written notice by either the Company or the Investor to the other party or parties, at any time after the date that is two hundred (200) calendar days after the Announcement Date (the "Termination Date") if the Closing shall not have occurred on or prior to the Termination Date; provided, that the right to terminate this Agreement under this Section 7.1(c) shall not be available to such party if the action or inaction of such party has been a principal cause of or resulted in the failure of the Closing to occur on or before the Termination Date and such action or failure to act constitutes a breach of this Agreement;
- (d) By either the Company or the Investor if a Governmental Entity shall have issued a non-appealable final order, decree or ruling or taken any other action having the effect of permanently restraining, enjoining or otherwise prohibiting the consummation of the Transaction;
- (e) By the Investor at any time prior to the Closing, if (i) the Company is in breach of a Major Representation or a Major Undertaking (ii) such breach is not cured or capable of being cured by the earlier of the Termination Date and twenty (20) days following written notice of such breach from the Investor (to the extent such breach is curable) and (iii) such breach, if not cured, would render the conditions set forth in Section 6.3 incapable of being satisfied; or
- (f) By the Company at any time prior to the Closing, if (i) the Investor in breach of their representations, warranties or covenants in this Agreement, (ii) such breach is not cured or capable of being cured by the earlier of the Termination Date and twenty (20) days following written notice of such breach from the Company (to the extent such breach is curable) and (iii) such breach, if not cured, would render the conditions set forth in Section 6.2 incapable of being satisfied.
- 7.2 Effect of Termination. In the event of the termination of this Agreement as provided in Section 7.1, all obligations and agreements of the parties set forth in this Agreement shall become void and of no further force and effect whatsoever and there shall be no liability or obligation on the part of the parties hereto except (i) for any liability of the Company as a result of a breach of this Agreement prior to such termination, and (ii) that the provisions of this Section 7.2, Section 5.2, and ARTICLE 8 (and all related

definitions in <u>ARTICLE 1</u>) shall survive any such termination and shall be enforceable hereunder.

ARTICLE 8

General Provisions

- 8.1 Amendments and Waivers. This Agreement may not be amended except by an instrument in writing signed on behalf of each of the parties hereto. Except as otherwise provided in this Agreement, any failure of any party to comply with any obligation, covenant, agreement or condition herein may be waived by the party entitled to the benefits thereof only by a written instrument signed by the party granting such waiver, but such waiver shall not operate as a waiver of, or estoppel with respect to, any subsequent or other failure. The failure of any party to this Agreement to assert any of its rights under this Agreement or otherwise shall not constitute a waiver of such rights.
- Assignment. This Agreement and the rights and obligations hereunder shall not be assignable or transferable by any party without the prior written consent of the other parties hereto; provided, that, prior to the Closing, the BDT Fund shall be permitted to assign a portion of its obligation to contribute the Fund Commitment Amount (the "Assigned Amount") pursuant to Section 5.11 to one or more limited partners of the BDT Fund (a "Permitted Assignee"), subject to the satisfaction of the following conditions: (a) such limited partner shall not be a Competitor of the Company, and (b) in connection with such assignment, the BDT Fund and such limited partner(s) shall execute and deliver instruments of transfer and assignment, under the terms of which the Permitted Assignee will undertake to pay to the Investor an amount equal to the Assigned Amount, and shall provide "certain funds" supporting documentation to the Company and Cash Confirmation Advisor and its counsel, in each case, in form and substance satisfactory to the Company and Cash Confirmation Advisor and its counsel but no more onerous than that provided by the BDT Fund; provided, further, that, following the Closing Date, the Investor may designate one or more Permitted Transferees to acquire a portion or all of the Purchased Shares, and to become a party to the Registration Rights Agreement and to have the rights of the Investor with respect to the Purchased Shares contained herein, upon such Permitted Transferee making the same representations and warranties to the Company as are set forth in Section 4.3 hereof, and agreeing to be bound (pursuant to an instrument that is reasonably satisfactory to the Company) by the post-closing covenants of the Investor set forth herein that would apply to such Purchased Shares. Any attempted assignment in violation of this Section 8.2 shall be void. Notwithstanding any assignment pursuant to this Section 8.2, prior to the Closing no Permitted Transferee shall be permitted to exercise any rights of the Investor under this Agreement (including any right to terminate the Agreement or to invoke any conditions precedent set forth in Section 6).
- 8.3 No Third-Party Beneficiaries. This Agreement is for the sole benefit of the parties hereto and their permitted assigns and nothing herein expressed or implied shall give or be construed to give to any person, other than the parties hereto and such permitted assigns, any legal or equitable rights hereunder.

8.4 Notices. All notices or other communications required or permitted to be given hereunder shall be in writing and shall be delivered by hand or sent by a nationally recognized overnight courier service (with tracking capability), and shall be deemed given when received, as follows:

if to the Investor, to:

401 North Michigan, Suite 3100 Chicago, Illinois 60611 Attention: William Bush Fax: (312) 832-1700

with a copy (which shall not constitute notice) to:

Munger, Tolles & Olson LLP 355 South Grand Avenue, 35th Floor Los Angeles, California 90071 Attention: Mary Ann Todd & Brett Rodda Fax: (213) 687-3702

if to the Company, to:

Colfax Corporation
8170 Maple Lawn Blvd., Suite 180
Fulton, Maryland 20759
Attention: Lynne Puckett
Fax: (301) 323-9001
with a copy (which shall not constitute notice) to:

Skadden, Arps, Slate, Meagher & Flom LLP 40 Bank Street London E14 5DS United Kingdom Attention: Scott V. Simpson & James A. McDonald Fax: (44) 20 70 72 7000

- **8.5** Counterparts. This Agreement may be executed in one or more counterparts, all of which shall be considered one and the same agreement, and shall become effective when one or more such counterparts have been signed by each of the parties hereto and delivered to the other parties hereto.
- 8.6 Entire Agreement. This Agreement and the Ancillary Agreements, along with the Schedules and the Exhibits hereto and thereto, contain the entire agreement and understanding among the parties hereto with respect to the subject matter hereof and thereof and supersede all prior agreements and understandings relating to such subject matter. None of the parties shall be liable or bound to any other party in any manner by any representations, warranties or covenants relating to such subject matter except as specifically set forth herein or in the Ancillary Agreements.

- 8.7 Interpretation; Exhibits and Schedules. The headings contained in this Agreement, in any Exhibit or Schedule hereto and in the table of contents to this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement. All Exhibits and Schedules annexed hereto or referred to herein are hereby incorporated in and made a part of this Agreement as if set forth in full herein. Any capitalized terms used in any Schedule or Exhibit but not otherwise defined therein shall have the meaning assigned to such term in this Agreement. When a reference is made in this Agreement to a Section, Exhibit or Schedule, such reference shall be to a Section of, or an Exhibit or Schedule to, this Agreement unless otherwise indicated.
- 8.8 Severability. If any provision of this Agreement (or any portion thereof) or the application of any such provision (or any portion thereof) to any person or circumstance shall be held invalid, illegal or unenforceable in any respect by a court of competent jurisdiction, such invalidity, illegality or unenforceability shall not affect any other provision hereof (or the remaining portion thereof) or the application of such provision to any other persons or circumstances, so long as the economic and legal substance of the Transaction is not affected in a manner materially adverse to any party hereto.
- 8.9 Consent to Jurisdiction. Each of the parties hereby consents to the exclusive jurisdiction of the state and federal courts sitting in the City of New York in any action on a claim arising out of, under or in connection with this Agreement or the Transaction.
- **8.10** Governing Law. This Agreement shall be governed and construed in accordance with the laws of the State of New York.
- 8.11 Specific Performance. The parties agree that irreparable damage would occur if any provision of this Agreement were not performed in accordance with its terms and that each party shall be entitled to an injunction or injunctions to prevent breaches of this Agreement or to enforce specifically the performance of the terms and provisions hereof. Subject to the right to recover attorneys' fees pursuant to Section 8.13, unless and until the Closing occurs, such injunctive relief shall be the sole and exclusive remedy of the Company for any breach of this Agreement by the Investor or the BDT Fund.
- 8.12 Waiver of Jury Trial. Each party hereby waives, to the fullest extent permitted by Applicable Law, any right it may have to a trial by jury in respect to any litigation directly or indirectly arising out of, under or in connection with this Agreement, any Ancillary Agreement or the Transaction. Each party (a) certifies that no representative, agent or attorney of any other party has represented, expressly or otherwise, that such other party would not, in the event of litigation, seek to enforce the foregoing waiver and (b) acknowledges that it and the other parties hereto have been induced to enter into this Agreement and the Ancillary Agreements, as applicable, by, among other things, the mutual waivers and certifications in this Section 8.12.
- 8.13 Attorneys' Fees. In the event any Action is brought in respect of this Agreement or the Transaction by the parties to this Agreement or their Affiliates, the prevailing party in such Action shall be entitled to recover reasonable attorneys' fees and

other costs incurred in such Action, in addition to any relief to which such party may be entitled.

- 8.14 No Personal Liability of Partners, Directors, Officers, Owners, Etc. No director, officer, employee, incorporator, stockholder, managing member, member, general partner, limited partner, principal or other agent of the Investor or the Main Fund shall have any liability for any obligations of the Investor or the Main Fund under this Agreement or for any claim based on, in respect of, or by reason of, the obligations of the Investor or the Main Fund hereunder. The Company waives and releases all such liability. This waiver and release is a material inducement to the Investor's and the Main Fund's (solely in the case of Section 5.11) entry into this Agreement. No director, officer, employee, incorporator, stockholder, managing member, member, general partner, limited partner, principal or other agent of the Company shall have any liability for any obligations of the Company under this Agreement or for any claim based on, in respect of, or by reason of, the obligations of the Company hereunder. The Investor waives and releases all such liability. This waiver and release is a material inducement to the Company's entry into this Agreement.
- 8.15 Rights of Holders. Each party to this Agreement shall have the absolute right to exercise or refrain from exercising any right or rights that such party may have by reason of this Agreement, including the right to consent to the waiver or modification of any obligation under this Agreement, and such party shall not incur any liability to any other party or other holder of any securities of the Company as a result of exercising or refraining from exercising any such right or rights.
- 8.16 Adjustment in Share Numbers and Prices. In the event of any (i) stock split, (ii) subdivision, (iii) dividend or distribution payable in shares of Common Stock (or other securities or rights convertible into or entitling the holder thereof to receive directly or indirectly shares of Common Stock), (iv) combination or (v) other similar recapitalization or event, in each case, occurring after the date hereof and prior to the Closing Date, each reference in this Agreement and the Ancillary Agreements to a number of shares or a price per share shall be amended to appropriately account for such event.
- 8.17 Construction. The parties acknowledge that each party and its counsel have participated in the negotiation and preparation of this Agreement. This Agreement shall be construed without regard to any presumption or other rule requiring construction against the party causing the Agreement to be drafted. Each covenant, term and provision of this Agreement shall be construed simply according to its fair meaning and not strictly for or against any party hereto.

[Signature page follows.]

IN WITNESS WHEREOF, the Company, the Investor, the BDT Fund and the Designated Stockholders have executed this Agreement as of the date first above written.

COMPANY COLFAX CORPORATION By: Name: Title: DESIGNATED STOCKHOLDERS (Joining solely for purposes of Section 5.12) Mitchell P. Rales Steven M. Rales **INVESTOR** BDT CF ACQUISITION VEHICLE, LLC By: Name: Title: **BDT FUND** (Joining solely for purposes of Section 5.11) BDT CAPITAL PARTNERS FUND I, L.P. By: BDT CP GP I, LLC, its general partner by: BDT Capital Partners, LLC, its managing Member By: _ Name: Title: BDT CAPITAL PARTNERS FUND I-A, L.P. By: BDT CP GP I, LLC, its general partner By: BDT Capital Partners, LLC, its managing member

By: _

Name: Title: In WITNESS WHEREOF, the Company, the Investor, the BDT Fund and the Designated Stockholders have executed this Agreement as of the date first above written.

COMPANY

71-7
By: My melst class Namet A. LymeRickett
Namet A. LymePickett
True SVP. General Cansel and Secretary
DESIGNATED STOCKHOLDERS
(Joining solely for purposes of Section 5.12)
Mitchell P. Rales
Steven M. Rales
INVESTOR
BDT CF ACQUISITION VEHICLE, LLC
By:
Name: William R. Bush
Title: Vice President, General Counsel and Secretary
BDT FUND
(Joining solely for purposes of Section 5.11)
(Joining solely for purposes of Section 5.11) BDT CAPITAL PARTNERS FUND I, L.P., By: BDT CP GP I, LLC, its general partner
(Joining solely for purposes of Section 5.11) BDT CAPITAL PARTNERS FUND I, L.P.,
(Joining solely for purposes of Section 5.11) BDT CAPITAL PARTNERS FUND I, L.P., By: BDT CP GP I, LLC, its general partner
(Joining solely for purposes of Section 5.11) BDT CAPITAL PARTNERS FUND I, L.P., By: BDT CP GP I, LLC, its general partner By: BDT Capital Partners, LLC, its managing member By: Name: William R. Bush
(Joining solely for purposes of Section 5.11) BDT CAPITAL PARTNERS FUND I, L.P., By: BDT CP GP I, LLC, its general partner By: BDT Capital Partners, LLC, its managing member By:
(Joining solely for purposes of Section 5.11) BDT CAPITAL PARTNERS FUND I, L.P., By: BDT CP GP I, LLC, its general partner By: BDT Capital Partners, LLC, its managing member By: Name: William R. Bush
(Joining solely for purposes of Section 5.11) BDT CAPITAL PARTNERS FUND I, L.P., By: BDT CP GP I, LLC, its general partner By: BDT Capital Partners, LLC, its managing member By: William R. Bush Title: Vice President, General Counsel and Secretary BDT CAPITAL PARTNERS FUND I-A, L.P. By: BDT CP GP I, LLC, its general partner
(Joining solely for purposes of Section 5.11) BDT CAPITAL PARTNERS FUND I, L.P., By: BDT CP GP I, LLC, its general partner By: BDT Capital Partners, LLC, its managing member By: Name: William R. Bush Title: Vice President, General Counsel and Secretary BDT CAPITAL PARTNERS FUND I-A, L.P.
(Joining solely for purposes of Section 5.11) BDT CAPITAL PARTNERS FUND I, L.P., By: BDT CP GP I, LLC, its general partner By: BDT Capital Partners, LLC, its managing member By: Name: William R. Bush Title: Vice President, General Counsel and Secretary BDT CAPITAL PARTNERS FUND I-A, L.P. By: BDT CP GP I, LLC, its general partner By: BDT Capital Partners, LLC, its managing member
(Joining solely for purposes of Section 5.11) BDT CAPITAL PARTNERS FUND I, L.P., By: BDT CP GP I, LLC, its general partner By: BDT Capital Partners, LLC, its managing member By: William R. Bush Title: Vice President, General Counsel and Secretary BDT CAPITAL PARTNERS FUND I-A, L.P. By: BDT CP GP I, LLC, its general partner

IN WITNESS WHEREOF, the Company, the Investor, the BDT Fund and the Designated Stockholders have executed this Agreement as of the date first above written.

Colfax Corporation
By: Name: Title:
DESIGNATED STOCKHOLDERS (Joining solely for purposes of Section 5.12)
MitchelhP Rales
Steven M. Rales
INVESTOR
BDT CF ACQUISITION VEHICLE, LLC
By: Name: William R. Bush Title: Vice President, General Counsel and Secretary
BDT FUND (Joining solely for purposes of Section 5.11)
BDT CAPITAL PARTNERS FUND I, L.P.,
By: BDT CP GP I, LLC, its general partner By: BDT Capital Partners, LLC, its managing member
By: Name: William R. Bush Title: Vice President, General Counsel and Secretary
BDT CAPITAL PARTNERS FUND I-A, L.P.
By: BDT CP GP I, LLC, its general partner By: BDT Capital Partners, LLC, its managing member
By:Name: William R. Bush

Title: Vice President, General Counsel and Secretary

COMPANY

IN WITNESS WHEREOF, the Company, the Investor, the BDT Find and the Designated Stockholders have executed this Agreement as of the date first above written COMPANY Colfax Corporation Бу: Name: Title: DESIGNATEDSTOCKHOLDERS (Joining solely for purposes of Section 5.12) Mitchell P. Reles INVESTOR BDTCFACQUISITIONVEHICLE,LLC By: Name: Title: BDT FUND (Joining solely for purposes of Section 5.11) BDT CAPITAL PARTNERS FUND L L.P. By: Name: Title: BDT CAPITAL PARTNERS FUND I-A, L P. By: Name:

Title:

IN WITNESS WHEREOF, the Company, the Investor, the BDT Fund and the Designated Stockholders have executed this Agreement as of the date first above written.

Designated	Stockholders	have e	executed	this.	Agreement	as	of the da	te first	above	written
COMPAN	v									

Ву:
Name: Title:
DESIGNATED STOCKHOLDERS (Joining solely for purposes of Section 5.12)
Mitchell P. Rales
Steven M. Rales
INVESTOR
BDT CF ACQUISITION VEHICLE, LLC
By: Name: William R. Bush Title: Vice President, General Counsel and Secretary
BDT FUND (Joining solely for purposes of Section 5.11)
BDT CAPITAL PARTNERS FUND I, L.P.,
By: BDT CP GP I, LLC, its general partner By: BDT Capital Partners, LLC, its managing member
By: William R. Bask
Name: William R. Bush Title: Vice President, General Counsel and Secretary
BDT CAPITAL PARTNERS FUND I-A, L.P.
By: BDT CP GP I, LLC, its general partner By: BDT Capital Partners, LLC, its managing member
Dy.
Name: William R. Bush Title: Vice President, General Counsel and Secretary

Colfax Corporation

COMPANY DISCLOSURE SCHEDULE

(the "Disclosure Schedule")

This Disclosure Schedule is being furnished by Colfax Corporation (the "Company") in connection with the execution and delivery of the Securities Purchase Agreement (the "Agreement"), delivered concurrently herewith, among the Company and BDT CF Acquisition Vehicle, LLC. Unless the context otherwise requires, all capitalized terms used in this Disclosure Schedule shall have the respective meanings assigned to them in the Agreement.

This Disclosure Schedule is qualified in its entirety by reference to the specific provisions in the Agreement. The headings contained in this Disclosure Schedule are included for convenience only and are not intended to limit the effect of the disclosures contained herein or to expand the scope of the information required to be disclosed herein. The schedule numbers in this Disclosure Schedule correspond to the section numbers in the Agreement, provided, that any information disclosed herein under any section shall be deemed to be disclosed with respect to any other section of the Agreement to which the relevance of such item is reasonably apparent from the face of such disclosure.

Schedule 3.3(c)

Capital Structure

Equity awards and grants made pursuant to the terms of the Company's 2008 Omnibus Incentive Plan and Director compensation program, including the Director Deferred Compensation Plan, as outstanding in the SEC Report for the three months ended June 30, 2011 and any awards under the plan granted in the ordinary course of business since such SEC Report.

Schedule 3.3 (d)

Capital Structure

Equity awards and grants made pursuant to the terms of the Company's 2008 Omnibus Incentive Plan and Director compensation program, including the Director Deferred Compensation Plan, as outstanding in the SEC Report for the three months ended June 30, 2011 and any awards under the plan granted in the ordinary course of business since such SEC Report.

Schedule 3.6(a)

SEC Reports and Company Financial Statements

The Company's Report on Form 10-Q for the Quarter ended October 1, 2010, which was non-timely filed on December 13, 2010.